%2multibyte Version: 5.50.0.2890 CodePage: 1253
%\input{tcilatex}
%\input{tcilatex}
%\input{tcilatex}


\documentclass{amsart}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{amsthm,amsfonts,amsmath,amssymb}
\usepackage{geometry}

\setcounter{MaxMatrixCols}{10}
%TCIDATA{OutputFilter=LATEX.DLL}
%TCIDATA{Version=5.50.0.2890}
%TCIDATA{Codepage=1253}
%TCIDATA{<META NAME="SaveForMode" CONTENT="1">}
%TCIDATA{BibliographyScheme=Manual}
%TCIDATA{LastRevised=Friday, May 25, 2018 16:50:44}
%TCIDATA{<META NAME="GraphicsSave" CONTENT="32">}

\geometry{left=3cm,right=3cm,top=2.5cm,bottom=2.5cm}
\newtheorem{theorem}{Theorem}
\newtheorem{corollary}{Corollary}
\newtheorem{definition}{Definition}
\newtheorem{lemma}{Lemma}
\newtheorem{remark}{Remark}
\numberwithin{equation}{section}


\begin{document}
\begin{thebibliography}{99}
\bibitem{alomari} M. Alomari and \ M. Darus: \textit{The Hadamards
inequality for }$s$\textit{-convex function of }$2$\textit{-variables on the
coordinates}. Int. J. Math. Anal. 2(13), 629--638 (2008).

\bibitem{alomari2} M. Alomari andM. Darus, \textit{Fej\'{e}r inequality for
double integrals}, Facta Universitatis (NI\v{S}), Ser. Math. Inform. 24
(2009), 15-28.

\bibitem{alomari3} M. Alomari, M. Darus and S.S. Dragomir, \textit{New
inequalities of Simpson\'{\i}s type for }$s$\textit{-convex functions with
applications}, RGMIA Res. Rep. Coll., 12 (4) (2009), Article 9.

\bibitem{bakula} M. K. Bakula, \textit{An improvement of the
Hermite-Hadamard inequality for functions convex on the coordinates},
Australian journal of mathematical analysis and applications, 11(1) (2014),
1-7.

\bibitem{budak} H. Budak and M. Z. Sar\i kaya, \textit{Hermite-Hadamard-Fej%
\'{e}r inequalities for double integrals}, 2018, ResearchGate Article,
10.13140/RG.2.2.14525.97760.

\bibitem{chen} F.Chen, \textit{A note on the Hermite-Hadamard inequality for
convex functions on the co-ordinates}, J. Math. Inequal., 8(4), (2014),
915-923.

\bibitem{dragomir} S.S. Dragomir, \textit{On Hadamards inequality for convex
functions on the co-ordinates in a rectangle from the plane}. Taiwan. J.
Math. 4, 775--788 (2001).

\bibitem{Dragomir1} S. S. Dragomir and C. E. M. Pearce, \textit{Selected
Topics on Hermite-Hadamard Inequalities and Applications}, RGMIA Monographs,
Victoria University, 2000.

\bibitem{dragomir5} S.S. Dragomir, R.P. Agarwal and P. Cerone, \textit{On
Simpson\'{\i}s inequality and applications}, J. of Inequal. Appl., 5(2000),
533-579.

\bibitem{dragomir7} S.S. Dragomir, \textit{On Simpson's quadrature formula
for Lipschitzian mappings and applications} Soochow J. Mathematics, 25
(1999), 175--180.

\bibitem{du} T. Du, Y. Li \ and Z. Yang, \textit{A generalization of
Simpson's inequality via differentiable mapping using extended }$(s,m)$%
\textit{-convex functions, }Applied Mathematics and Computation 293 (2017)
358--369

\bibitem{farid} G. Farid, M. Marwan and Atiq Ur Rehman, \textit{%
Fejer-Hadamard inequlality for convex functions on the co-ordinates in a
rectangle from the plane}, International Journal of Analysis and
Applications10(1), (2016), 40-47.

\bibitem{Fejer} L. Fejer, \textit{\"{U}ber die Fourierreihen, }II. Math.
Naturwiss. Anz Ungar. Akad. Wiss., 24 (1906), 369--390. (Hungarian).

\bibitem{hussain} S. Hussain and S. Qaisar, \textit{More results on
Simpson's type inequality through convexity for twice differentiable
continuous mappings}. Springer Plus (2016), 5:77.

\bibitem{kav} H. Kavurmaci, A. O. Akdemir, E. Set and M. Z. Sarikaya,
\textit{Simpson's type inequalities for }$m-$\textit{\ and }$(\alpha ,m)$%
\textit{-geometrically convex functions}, Konuralp Journal of Mathematics,
2(1), pp:90-101, 2014.

\bibitem{lat} M. A. Latif, S. Hussain and S. S. Dragomir, \textit{On some
new Fejer-type inequalities for coordinated convex functions}, TJMM, 3
(2011), No. 2, 57-80.

\bibitem{latif} M. A. Latif, On some \textit{Fejer-type inequalities for
double integrals}, Tamkang Journal of Mathematics, 43(3), 2012, 423-436.

\bibitem{latif2} M. A. Latif, S. S. Dragomir, and E. Momoniat, \textit{%
Weighted generalization of some integral inequalities for differentiable
co-ordinated convex functions}, Politehn. Univ. Bucharest Sci. Bull. Ser. A
Appl. Math. Phys. 78 (2016), no. 4, 197--210.

\bibitem{latif4} M. A. Latif, S. S. Dragomir, and E. Momoniat, \textit{%
Generalization of some Inequalities for differentiable co-ordinated convex
functions with applications}, Moroccan J. Pure and Appl. Anal. 2(1), 2016,
12--32.

\bibitem{latif3} LM. A. Latif and S. S. Dragomir, \textit{On some new
inequalities for differentiable co-ordinated convex functions}, J.Inequal.
Appl., 2012, (2012): 28.

\bibitem{liu} B.Z. Liu, \textit{An inequality of Simpson type}, Proc. R.
Soc. A, 461 (2005), 2155-2158.

\bibitem{ozdemir} M.E. Ozdemir, C. Yildiz and A.O. Akdemir, \textit{On the
co-ordinated convex functions,} Appl. Math. Inf. Sci. 8(3), 1085--1091
(2014).

\bibitem{ozdemir2} M. E. Ozdemir, A. O. Akdemir and H. Kavurmac\i , \textit{%
On the Simpson's inequality for convex functions on the co-ordinates},
Turkish Journal of Analysis and Number Theory. 2014, 2(5), 165-169.

\bibitem{Pecaric} J.E. Pe\v{c}ari\'{c}, F. Proschan and Y.L. Tong, \textit{%
Convex Functions, Partial Orderings and Statistical Applications}, Academic
Press, Boston, 1992.

\bibitem{pecaric2} J. Pecaric., and S. Varosanec, \textit{A note on
Simpson's inequality for functions of bounded variation}, Tamkang Journal of
Mathematics, Volume 31, Number 3, Autumn (2000), 239--242.

\bibitem{qasiar} S. Qaisar, C.J. He, S. Hussain, \textit{A generalizations
of Simpson's type inequality for differentiable functions using }$\mathit{%
(\alpha ,m)}$\textit{-convex functions and applications}, J. Inequal. Appl.
2013 (2013) 13. Article 158.

\bibitem{sarikaya} M. Z. Sarikaya, E. Set, M. E. Ozdemir and S. S. Dragomir%
\textit{,New some Hadamard's type inequalities for co-ordinated convex
functions,} Tamsui Oxford Journal of Information and Mathematical Sciences,
28(2) (2012) 137-152.

\bibitem{sarikaya1} M. Z. Sarikaya, E. Set and M. E. Ozdemir, \textit{On new
inequalities of Simpson's type for }$\mathit{s}$\textit{-convex functions},
Computers and Mathematics with Applications 60 (2010) 2191--2199.

\bibitem{sarikaya2} M.Z. Sarikaya, E. Set, M.E. \"{O}zdemir, On new
inequalities of Simpson's type for convex functions, RGMIA Res. Rep. Coll.
13 (2) (2010) Article2.

\bibitem{sarikaya3} M. Z.Sarikaya, E. Set and M. E. Ozdemir, On new
inequalities of Simpson's type for functions whose second derivatives
absolute values are convex, Journal of Applied Mathematics, Statistics and
Informatics , 9 (2013), No. 1.

\bibitem{sarikaya4} M.Z. Sar\i kaya, T. Tun\c{c} and H. Budak, \textit{%
Simpson's type inequality for }$\mathit{F}$\textit{-convex function}, Facta
Universitatis Ser. Math. Inform., 32(5), (2017), 747--753.

\bibitem{set} E. Set, M.E. \"{O}zdemir, S.S. Dragomir, \textit{On the
Hermite-Hadamard inequality and other integral inequalities involving two
functions}, J. Inequal. Appl. (2010) 9. Article ID 148102.

\bibitem{set1} E. Set, M. E. Ozdemir and M. Z. Sarikaya, \textit{On new
inequalities of Simpson's type for quasi-convex functions with applications}%
, Tamkang Journal of Mathematics, 43 (2012), no. 3, 357--364.

\bibitem{set2} E. Set, M. Z. Sarikaya and N. Uygun, \textit{On new
inequalities of Simpson's type for generalized quasi-convex functions},
Advances in Inequalities and Applications, 2017, 2017:3, pp:1-11.

\bibitem{tseng2} K. L. Tseng, G. S. Yang and S.S. Dragomir, \textit{On
weighted Simpson type inequalities and applications} Journal of mathematical
inequalities, Vol. 1, number 1 (2007), 13--22.

\bibitem{wang} D.Y. Wang, K.L. Tseng and G. S. Yang, \textit{Some Hadamard's
inequalities for co-ordinated convex functions in a rectangle from the plane}%
. Taiwan. J. Math. 11, 63--73 (2007).

\bibitem{ujevic} N. Ujevic,\textit{\ Double integral inequalities of Simpson
type and applications}, J. Appl. Math. Comput., 14 (2004), no:1-2, p.
213-223.

\bibitem{xi} B.Y. Xi, J. Hua and F. Qi, \textit{Hermite-Hadamard type
inequalities for extended }$s$\textit{-convex functions on the co-ordinates
in a rectangle.} J. Appl. Anal. 20(1), 1--17 (2014).

\bibitem{xi2} R. Xiang and F. Chen, \textit{On some integral inequalities
related to Hermite-Hadamard-Fej\'{e}r inequalities for coordinated convex
functions}, Chinese Journal of Mathematics, Volume 2014, Article ID 796132,
10 pages

\bibitem{yang3} Z.Q. Yang, Y.J. Li andT. Du, \textit{A generalization of
Simpson type inequality via differentiable functions using }$(s,m)$\textit{%
-convex functions}, Ital. J. Pure Appl. Math. 35 (2015) 327--338

\bibitem{yil} M. E. Y\i ld\i r\i m, A. Akkurt and H. Y\i ld\i r\i m, \textit{%
Hermite-Hadamard type inequalities for co-ordinated }$\left( \alpha
_{1},m_{1}\right) -\left( \alpha _{2},m_{2}\right) $\textit{-convex
functions via fractional integrals}, Contemporary Analysis and Applied M
athematics, 4(1), 48-63, 2016.
\end{thebibliography}

\title[Weighted Hermite-Hadamard and Simpson inequalities...]{Weighted
Hermite-Hadamard and Simpson type inequalities for double integrals}
\address{Department of Mathematics, \ Faculty of Science and Arts, D\"{u}zce
University, D\"{u}zce-TURKEY}
\author{H\"{u}seyin BUDAK}
\email{hsyn.budak@gmail.com}
\author{Fatma Ertu\u{g}ral}
\address{Department of Mathematics, Faculty of Science and Arts, D\"{u}zce
University, D\"{u}zce, Turkey}
\email{fatmaertugral14@gmail.com}
\author{Mehmet Zeki SARIKAYA}
\address{Department of Mathematics, \ Faculty of Science and Arts, D\"{u}zce
University, D\"{u}zce-TURKEY}
\email{sarikayamz@gmail.com}
\keywords{\textbf{\thanks{\textbf{2010 Mathematics Subject Classification.}
26D07, 26D10, 26D15, 26B15, 26B25.} }Hermite-Hadamard-Fejer inequality,
Simpson inequality, co-ordinated convex, integral inequalities.}

\begin{abstract}
In this paper, we first obtain two weighted identities for twice partially
differentiable mappings. Moreover, utilizing these equalities, we establish
the weighted Herrmite-Hadamard type inequalities and weighted Simpson type
inequalities for co-ordinated convex functions in a rectangle from the plane
$%
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
^{2}$, respectivelly. The results given in this paper provide
generalizations of some result established in earlier works.
\end{abstract}

\maketitle

\section{Introduction}

The Hermite-Hadamard inequality discovered by C. Hermite and J. Hadamard
see, e.g., \cite{Dragomir1}, \cite[p.137]{Pecaric}) is one of the most well
established inequalities in the theory of convex functions with a
geometrical interpretation and many applications. These inequalities state
that if $f:I\rightarrow \mathbb{R}$ is a convex function on the interval $I$
of real numbers and $a,b\in I$ with $a<b$, then
\begin{equation}
f\left( \frac{a+b}{2}\right) \leq \frac{1}{b-a}\int
\limits_{a}^{b}f(x)dx\leq \frac{f\left( a\right) +f\left( b\right) }{2}.
\label{E1}
\end{equation}%
Both inequalities hold in the reversed direction if $f$ is concave. We note
that Hermite-Hadamard inequality may be regarded as a refinement of the
concept of convexity and it follows easily from Jensen's inequality.
Hermite-Hadamard inequality for convex functions has received renewed
attention in recent years and a remarkable variety of refinements and
generalizations have been studied.

The weighted version of the inequalities (\ref{E1}), so-called
Hermite-Hadamard-Fej\'{e}r inequalities, was given by Fejer in \cite{Fejer}
as follow:

\begin{theorem}
$f:[a,b]\rightarrow \mathbb{R}$, be a convex function, then the inequality%
\begin{equation}
f\left( \frac{a+b}{2}\right) \int \limits_{a}^{b}g(x)dx\leq \int
\limits_{a}^{b}f(x)g(x)dx\leq \frac{f(a)+f(b)}{2}\int \limits_{a}^{b}g(x)dx
\label{H1}
\end{equation}%
holds, where $g:[a,b]\rightarrow \mathbb{R}$ is nonnegative, integrable, and
symmetric about $x=\frac{a+b}{2}$ (i.e. $g(x)=g(a+b-x)$)$.$
\end{theorem}

Onthe other hand, the following inequality is well known in the literature
as Simpson's inequality.

\begin{theorem}
Let $\ f:\left[ a,b\right] \rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ be a four times continuously differentiable mapping on $(a,b)$ and $%
\left
\Vert f^{(4)}\right \Vert _{\infty }=\sup \left \vert
f^{(4)}(x)\right \vert <\infty .$Then, the following inequality holds:
\end{theorem}

\begin{equation*}
\left \vert \frac{1}{3}\left[ \frac{f(a)+f(b)}{2}+2f\left( \frac{a+b}{2}%
\right) \right] -\frac{1}{b-a}\int_{a}^{b}f\left( x\right) dx\right \vert
\leq \frac{1}{2880}\left \Vert f^{(4)}\right \Vert _{\infty }(b-a)^{4}.
\end{equation*}

For recent refinements, counterparts, generalizations and new Simpson's type
inequalities, see (\cite{alomari3}, \cite{dragomir5}-\cite{du}, \cite%
{hussain}, \cite{kav}, \cite{liu},\  \cite{pecaric2}, \cite{qasiar}, \cite%
{sarikaya1}-\cite{sarikaya4}, \cite{set1}, \cite{set2}, \cite{tseng2}, \cite%
{ujevic}, \cite{yang3}).

A formal defination for co-ordinated convex function may be stated as
follows:

\begin{definition}
A function $f:\Delta \rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ is called co-ordinated convex on $\Delta ,$ for all $(x,u),(y,v)\in \Delta
$ and $t,s\in \lbrack 0,1]$, if it satifies the following inequality:%
\begin{eqnarray}
&&f(tx+(1-t)\text{ }y,su+(1-s)\text{ }v)  \label{T4} \\
&&  \notag \\
&\leq &ts\text{ }f(x,u)+t(1-s)f(x,v)+s(1-t)f(y,u)+(1-t)(1-s)f(y,v).  \notag
\end{eqnarray}
\end{definition}

The mapping $f$ is a co-ordinated concave on $\Delta $ if the inequality (%
\ref{T4}) holds in reversed direction for all $t,s\in \lbrack 0,1]$ and $%
(x,u),(y,v)\in \Delta $.

In \cite{dragomir}, Dragomir proved the following inequalities which is
Hermite-Hadamard type inequalities for co-ordinated convex functions on the
rectangle from the plane $%
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
^{2}.$

\begin{theorem}
Suppose that $f:\Delta \rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ is co-ordinated convex, then we have the following inequalities:%
\begin{eqnarray}
f\left( \frac{a+b}{2},\frac{c+d}{2}\right) &\leq &\frac{1}{2}\left[ \frac{1}{%
b-a}\int \limits_{a}^{b}f\left( x,\frac{c+d}{2}\right) dx+\frac{1}{d-c}\int
\limits_{c}^{d}f\left( \frac{a+b}{2},y\right) dy\right]  \notag \\
&\leq &\frac{1}{(b-a)(d-c)}\int \limits_{a}^{b}\int \limits_{c}^{d}f(x,y)dydx
\label{E2} \\
&\leq &\frac{1}{4}\left[ \frac{1}{b-a}\int \limits_{a}^{b}f(x,c)dx+\frac{1}{%
b-a}\int \limits_{a}^{b}f(x,d)dx\right.  \notag \\
&+&\left. \frac{1}{d-c}\int \limits_{c}^{d}f(a,y)dy+\frac{1}{d-c}\int
\limits_{c}^{d}f(b,y)dy\right]  \notag \\
&&  \notag \\
&\leq &\frac{f(a,c)+f(a,d)+f(b,c)+f(b,d)}{4}.  \notag
\end{eqnarray}%
The above inequalities are sharp. The inequalities in (\ref{E2}) hold in
reverse direction if the mapping $f$ is a co-ordinated concave mapping.
\end{theorem}

Over the years, many papers are dedicated on the generalizations and new
versions of the inequalities (\ref{E2}) using the different type convex
functions. For the other Hermite-Hadamard type inequalities for co-ordinated
convex functions, please refer to (\cite{alomari}, \cite{alomari2}, \cite%
{bakula} ,\cite{chen}, \cite{ozdemir}, \cite{sarikaya}, \cite{set}, \cite%
{ozdemir}, \cite{wang}, \cite{xi}, \cite{yil})

In \cite{ozdemir2}, \"{O}zdemir et al. gave the following identity and using
the this identity, the authors established some Simpson type inequalities
for double integrals:

\begin{lemma}
\label{l1} $f:\Delta :=\left[ a,b\right] \times \left[ c,d\right]
\rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ be a twice partially differentiable mapping on $\Delta ^{\circ }$. If $%
\dfrac{\partial ^{2}f}{\partial t\partial s}\in L(\Delta ),$ then we have
the following equality%
\begin{eqnarray*}
&&\frac{f\left( a,\frac{c+d}{2}\right) +f\left( b,\frac{c+d}{2}\right)
+4f\left( \frac{a+b}{2},\frac{c+d}{2}\right) +f\left( \frac{a+b}{2},c\right)
+f\left( \frac{a+b}{2},d\right) }{9} \\
&&+\frac{f\left( a,c\right) +f\left( b,c\right) +f\left( a,d\right) +f\left(
b,d\right) }{36} \\
&&-\frac{1}{6\left( b-a\right) }\int_{a}^{b}\left[ f\left( x,c\right)
+4f\left( x,\frac{c+d}{2}\right) +f\left( x,d\right) \right] dx \\
&&-\frac{1}{6\left( d-c\right) }\int_{c}^{d}\left[ f\left( a,y\right)
+4f\left( \frac{a+b}{2},y\right) +f\left( b,y\right) \right] dy \\
&&+\frac{1}{\left( b-a\right) \left( d-c\right) }\int_{a}^{b}\int_{c}^{d}f%
\left( x,y\right) dydx \\
&=&\left( b-a\right) \left( d-c\right) \int_{0}^{1}\int_{0}^{1}q\left(
t,s\right) \frac{\partial ^{2}f}{\partial t\partial s}\left( at+\left(
1-t\right) b,cs+\left( 1-s\right) d\right) dsdt
\end{eqnarray*}%
which the mapping $q:\left[ 0,1\right] \times \left[ 0,1\right] \rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ is defined by%
\begin{equation*}
q\left( t,s\right) =\left \{
\begin{array}{ccc}
\left( t-\frac{1}{6}\right) \left( s-\frac{1}{6}\right) & 0\leq t\leq \frac{1%
}{2}, & 0\leq s\leq \frac{1}{2} \\
&  &  \\
\left( t-\frac{1}{6}\right) \left( s-\frac{5}{6}\right) & 0\leq t\leq \frac{1%
}{2}, & \frac{1}{2}\leq s\leq 1 \\
&  &  \\
\left( t-\frac{5}{6}\right) \left( s-\frac{1}{6}\right) & \frac{1}{2}\leq
t\leq 1, & 0\leq s\leq \frac{1}{2} \\
&  &  \\
\left( t-\frac{5}{6}\right) \left( s-\frac{5}{6}\right) & \frac{1}{2}\leq
t\leq 1, & \frac{1}{2}\leq s\leq 1.%
\end{array}%
\right.
\end{equation*}
\end{lemma}

Budak and Sar\i kaya proved the following Hermite-Hadamard-Fej\'{e}r
inequalities for double integrals in \cite{budak}:

\begin{theorem}
Let $p:\Delta :=\left[ a,b\right] \times \left[ c,d\right] \rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ be a positive, integrable and symmetric about $\frac{a+b}{2}$ and $\frac{%
c+d}{2}.$ Let $f:\Delta \rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ be a co-ordinated convex on $\Delta $, then we have the following
Hermite-Hadamard-Fejer type inequality%
\begin{eqnarray*}
&&f\left( \frac{a+b}{2},\frac{c+d}{2}\right) \int \limits_{a}^{b}\int
\limits_{c}^{d}p(x,y)dydx \\
&\leq &\frac{1}{2}\int \limits_{a}^{b}\int \limits_{c}^{d}\left[ f\left( x,%
\frac{c+d}{2}\right) +f\left( \frac{a+b}{2},y\right) \right] p(x,y)dydx \\
&& \\
&\leq &\int \limits_{a}^{b}\int \limits_{c}^{d}f(x,y)p(x,y)dydx \\
&& \\
&\leq &\frac{1}{4}\int \limits_{a}^{b}\int \limits_{c}^{d}\left[
f(x,c)+f(x,d)+f(a,y)+f(b,y)\right] p(x,y)dydx \\
&& \\
&\leq &\frac{f(a,c)+f(a,d)+f(b,c)+f(b,d)}{4}\int \limits_{a}^{b}\int
\limits_{c}^{d}p(x,y)dydx.
\end{eqnarray*}
\end{theorem}

Moreover, Farid et al. established a weighted version of the inequalities (%
\ref{E2}) in \cite{farid}. Please see (\cite{lat}-\cite{latif3}, \cite{xi2})
for other papers focused on Hermite-Hadamard-Fej\'{e}r inequalities for
co-ordinated convex functions.

The aim of this paper is to establish some weighed generalizations of
Hermite-Hadamard and Simpson type integral inequalities. The results
presented in this paper provide extensions of those given in \cite{ozdemir2}
and \cite{sarikaya}.

\section{Weighted Hermite-Hadamard type Inequalities}

In this section, we first prove a weighted identity for twice partially
differentiable mapping. Then, using this identity, we established a weighted
Hermite-Hadamard type inequality for co-ordinated convex mapping.

\begin{lemma}
\label{l2} Let $p:\Delta :=\left[ a,b\right] \times \left[ c,d\right]
\rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ be a positive and integrable function on $\Delta $ and let $f:\Delta
\rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ be a twice partially differentiable mapping on $\Delta ^{\circ }$. If $%
\dfrac{\partial ^{2}f}{\partial t\partial s}\in L(\Delta ),$ then we have
the following equality%
\begin{eqnarray}
&&\frac{f(a,c)+f(a,d)+f(b,c)+f(b,d)}{4}\int \limits_{a}^{b}\int
\limits_{c}^{d}p(x,y)dydx  \label{e20} \\
&&  \notag \\
&&-\frac{1}{4}\int \limits_{a}^{b}\int \limits_{c}^{d}\left[
f(x,c)+f(x,d)+f(a,y)+f(b,y)\right] p(x,y)dydx+\int \limits_{a}^{b}\int
\limits_{c}^{d}f(x,y)p(x,y)dydx  \notag \\
&&  \notag \\
&=&\frac{\left( b-a\right) \left( d-c\right) }{4}\int \limits_{0}^{1}\int
\limits_{0}^{1}\left[ \int \limits_{U_{1}(t)}^{U_{2}(t)}\int
\limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right] \dfrac{\partial ^{2}f}{%
\partial t\partial s}\left( U_{2}(t),V_{2}(s)\right) dsdt  \notag
\end{eqnarray}%
where $U_{1}(t)=(1-t)a+tb,$ $U_{2}(t)=ta+(1-t)b,$ $V_{1}(s)=(1-s)c+sd$ and $%
V_{2}(s)=sc+(1-s)d.$
\end{lemma}

\begin{proof}
Integrating the by parts we have,

\begin{eqnarray}
&&\int \limits_{0}^{1}\int \limits_{0}^{1}\left[ \int%
\limits_{U_{1}(t)}^{U_{2}(t)}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv%
\right] \dfrac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) dsdt  \label{e21} \\
&&  \notag \\
&=&\int \limits_{0}^{1}\left \{ \left. \frac{1}{a-b}\left[
\int \limits_{U_{1}(t)}^{U_{2}(t)}\int
\limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right] \dfrac{\partial f}{\partial s}%
\left( U_{2}(t),V_{2}(s)\right) \right \vert _{0}^{1}\right.  \notag \\
&&\left. -\int \limits_{0}^{1}\left[ \int%
\limits_{V_{1}(s)}^{V_{2}(s)}p(U_{2}(t),v)dv+\int%
\limits_{V_{1}(s)}^{V_{2}(s)}p(U_{1}(t),v)dv\right] \dfrac{\partial f}{%
\partial s}\left( U_{2}(t),V_{2}(s)\right) dt\right \} ds  \notag \\
&=&\int \limits_{0}^{1}\left \{ \frac{1}{b-a}\left[ \int
\limits_{a}^{b}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right] \dfrac{%
\partial f}{\partial s}\left( a,V_{2}(s)\right) +\frac{1}{b-a}\left[ \int
\limits_{a}^{b}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right] \dfrac{%
\partial f}{\partial s}\left( b,V_{2}(s)\right) \right.  \notag \\
&&\left. -\int \limits_{0}^{1}\left[ \int%
\limits_{V_{1}(s)}^{V_{2}(s)}p(U_{2}(t),v)dv\right] \dfrac{\partial f}{%
\partial s}\left( U_{2}(t),V_{2}(s)\right) dt-\int \limits_{0}^{1}\left[
\int \limits_{V_{1}(s)}^{V_{2}(s)}p(U_{1}(t),v)dv\right] \dfrac{\partial f}{%
\partial s}\left( U_{2}(t),V_{2}(s)\right) dt\right \} ds  \notag \\
&=&\frac{1}{b-a}\left \{ \int \limits_{0}^{1}\left[ \int
\limits_{a}^{b}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right] \dfrac{%
\partial f}{\partial s}\left( a,V_{2}(s)\right) ds+\int \limits_{0}^{1}%
\left[ \int \limits_{a}^{b}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv%
\right] \dfrac{\partial f}{\partial s}\left( b,V_{2}(s)\right) ds\right \}
\notag \\
&&-\int \limits_{0}^{1}\int \limits_{0}^{1}\left[ \int%
\limits_{V_{1}(s)}^{V_{2}(s)}p(U_{2}(t),v)dv\right] \dfrac{\partial f}{%
\partial s}\left( U_{2}(t),V_{2}(s)\right) dtds-\int \limits_{0}^{1}\int
\limits_{0}^{1}\left[ \int \limits_{V_{1}(s)}^{V_{2}(s)}p(U_{1}(t),v)dv%
\right] \dfrac{\partial f}{\partial s}\left( U_{2}(t),V_{2}(s)\right) dtds
\notag \\
&=&\frac{1}{b-a}\left \{ I_{1}+I_{2}\right \} -I_{3}-I_{4}.  \notag
\end{eqnarray}%
Using again the intergration by parts, we obtain%
\begin{eqnarray}
I_{1} &=&\int \limits_{0}^{1}\left[ \int
\limits_{a}^{b}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right] \dfrac{%
\partial f}{\partial s}\left( a,V_{2}(s)\right) ds  \label{e22} \\
&=&\frac{1}{c-d}\left. \left[ \int
\limits_{a}^{b}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right] f\left(
a,V_{2}(s)\right) \right \vert _{0}^{1}  \notag \\
&&-\int \limits_{0}^{1}\left[ \int
\limits_{a}^{b}p(u,V_{2}(s))du+\int \limits_{a}^{b}p(u,V_{1}(s))du\right]
f\left( a,V_{2}(s)\right) ds  \notag \\
&=&\frac{1}{d-c}\left( \int \limits_{a}^{b}\int
\limits_{c}^{d}p(u,v)dudv\right) \left[ f\left( a,c\right) +f(a,d)\right]
\notag \\
&&-\int \limits_{0}^{1}\int \limits_{a}^{b}p(u,V_{2}(s))f\left(
a,V_{2}(s)\right) duds-\int \limits_{0}^{1}\int
\limits_{a}^{b}p(u,V_{1}(s))f\left( a,V_{2}(s)\right) duds  \notag \\
&=&\frac{1}{d-c}\left( \int \limits_{a}^{b}\int
\limits_{c}^{d}p(u,v)dudv\right) \left[ f\left( a,c\right) +f(a,d)\right] -%
\frac{2}{d-c}\int \limits_{a}^{b}\int \limits_{c}^{d}p(u,v)f(a,v)dudv,
\notag
\end{eqnarray}%
and similarly,%
\begin{eqnarray}
I_{2} &=&\int \limits_{0}^{1}\left[ \int
\limits_{a}^{b}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right] \dfrac{%
\partial f}{\partial s}\left( b,V_{2}(s)\right) ds  \label{e23} \\
&=&\left. \frac{1}{c-d}\left[ \int
\limits_{a}^{b}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right] f\left(
b,V_{2}(s)\right) \right \vert _{0}^{1}  \notag \\
&&-\int \limits_{0}^{1}\left[ \int
\limits_{a}^{b}p(u,V_{2}(s))dudv+\int \limits_{a}^{b}p(u,V_{1}(s))dudv\right]
f\left( b,V_{2}(s)\right) ds  \notag \\
&=&\frac{1}{d-c}\left( \int \limits_{a}^{b}\int
\limits_{c}^{d}p(u,v)dudv\right) \left[ f\left( b,c\right) +f(b,d)\right] -%
\frac{2}{d-c}\int \limits_{a}^{b}\int \limits_{c}^{d}p(u,v)f(b,v)dudv.
\notag
\end{eqnarray}%
Onthe other hand, we get
\begin{eqnarray}
I_{3} &=&\int \limits_{0}^{1}\int \limits_{0}^{1}\left[ \int%
\limits_{V_{1}(s)}^{V_{2}(s)}p(U_{2}(t),v)dv\right] \dfrac{\partial f}{%
\partial s}\left( U_{2}(t),V_{2}(s)\right) dtds  \label{e24} \\
&=&\int \limits_{0}^{1}\left \{ \left. \frac{1}{c-d}\left[
\int \limits_{V_{1}(s)}^{V_{2}(s)}p(U_{2}(t),v)dv\right] f\left(
U_{2}(t),V_{2}(s)\right) \right \vert _{0}^{1}\right.  \notag \\
&&\left. -\int \limits_{0}^{1}\left[
p(U_{2}(t),V_{2}(s))+p(U_{2}(t),V_{1}(s))\right] f\left(
U_{2}(t),V_{2}(s)\right) ds\right \} dt  \notag
\end{eqnarray}%
\begin{eqnarray*}
&=&\int \limits_{0}^{1}\left \{ \frac{1}{d-c}\left( \int
\limits_{c}^{d}p(U_{2}(t),v)dv\right) \left[ f\left( U_{2}(t),c\right)
+f\left( U_{2}(t),d\right) \right] \right. \\
&&\left. -\int \limits_{0}^{1}p(U_{2}(t),V_{2}(s))f\left(
U_{2}(t),V_{2}(s)\right) ds-\int \limits_{0}^{1}p(U_{2}(t),V_{1}(s))f\left(
U_{2}(t),V_{2}(s)\right) ds\right \} \\
&=&\frac{1}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)\left[ f\left( u,c\right) +f\left(
u,d\right) \right] dvdu \\
&&-\frac{2}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)f\left( u,v\right) dvdu
\end{eqnarray*}%
and similarly,
\begin{eqnarray}
I_{4} &=&\int \limits_{0}^{1}\int \limits_{0}^{1}\left[ \int%
\limits_{V_{1}(s)}^{V_{2}(s)}p(U_{1}(t),v)dv\right] \dfrac{\partial f}{%
\partial s}\left( U_{2}(t),V_{2}(s)\right) dtds  \label{e25} \\
&=&\frac{1}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)\left[ f\left( u,c\right) +f\left(
u,d\right) \right] dvdu  \notag \\
&&-\frac{2}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)f\left( u,v\right) dvdu.  \notag
\end{eqnarray}%
If we substitute the equalities (\ref{e22})-(\ref{e25}) in (\ref{e21}), we
obtain%
\begin{eqnarray}
&&\int \limits_{0}^{1}\int \limits_{0}^{1}\left[ \int%
\limits_{U_{1}(t)}^{U_{2}(t)}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv%
\right] \dfrac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) dsdt  \label{e26} \\
&=&\frac{1}{\left( b-a\right) \left( d-c\right) }\left( \int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)dudv\right) \left[ f\left(
a,c\right) +f(a,d)\right] -\frac{2}{\left( b-a\right) \left( d-c\right) }%
\int \limits_{a}^{b}\int \limits_{c}^{d}p(u,v)f(a,v)dudv  \notag \\
&&+\frac{1}{\left( b-a\right) \left( d-c\right) }\left( \int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)dudv\right) \left[ f\left(
b,c\right) +f(b,d)\right] -\frac{2}{\left( b-a\right) \left( d-c\right) }%
\int \limits_{a}^{b}\int \limits_{c}^{d}p(u,v)f(b,v)dudv  \notag \\
&&-\frac{1}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)\left[ f\left( u,c\right) +f\left(
u,d\right) \right] dvdu+\frac{2}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)f\left( u,v\right) dvdu  \notag \\
&&-\frac{1}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)\left[ f\left( u,c\right) +f\left(
u,d\right) \right] dvdu+\frac{2}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)f\left( u,v\right) dvdu  \notag \\
&=&\frac{1}{\left( b-a\right) \left( d-c\right) }\left( \int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)dudv\right) \left[ f\left(
a,c\right) +f(a,d)+f\left( b,c\right) +f(b,d)\right]  \notag \\
&&-\frac{2}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)\left[ ff\left( u,c\right) +f\left(
u,d\right) +(a,v)+f(b,v)\right] dvdu  \notag \\
&&+\frac{4}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}p(u,v)f\left( u,v\right) dvdu.  \notag
\end{eqnarray}%
If we multiply the equality (\ref{e26}) by $\frac{\left( b-a\right) \left(
d-c\right) }{4},$ then we obtain the desired result (\ref{e20}).
\end{proof}

\begin{remark}
If we choose $p(x,y)=1$ in Lemma \ref{l2}, then the Lemma \ref{l2} reduces
to Lemma 1 in \cite{sarikaya}.
\end{remark}

\begin{theorem}
Let $p:\Delta :=\left[ a,b\right] \times \left[ c,d\right] \rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ be a positive and integrable function on $\Delta $ and let $f:\Delta
\rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ be a twice partially differentiable mapping on $\Delta ^{\circ }$. If $%
\left \vert \dfrac{\partial ^{2}f}{\partial t\partial s}\right \vert ^{q},$ $%
q>1,$ is a co-ordinated convex function on $\Delta ,$ then we have the
following inequality%
\begin{eqnarray}
&&\left \vert \frac{f(a,c)+f(a,d)+f(b,c)+f(b,d)}{4}\int
\limits_{a}^{b}\int \limits_{c}^{d}p(x,y)dydx\right.  \label{e27} \\
&&  \notag \\
&&\left. -\frac{1}{4}\int \limits_{a}^{b}\int \limits_{c}^{d}\left[
f(x,c)+f(x,d)+f(a,y)+f(b,y)\right] p(x,y)dydx+\int \limits_{a}^{b}\int
\limits_{c}^{d}f(x,y)p(x,y)dydx\right \vert  \notag \\
&&  \notag \\
&=&\frac{\left( b-a\right) \left( d-c\right) }{4}\left( \int
\limits_{0}^{1}\int \limits_{0}^{1}\left( H(t,s)\right) ^{p}dsdt\right) ^{%
\frac{1}{p}}  \notag \\
&&\times \left[ \frac{\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s%
}\left( a,c\right) \right \vert ^{q}+\left \vert \tfrac{\partial ^{2}f}{%
\partial t\partial s}\left( a,d\right) \right \vert ^{q}+\left \vert \tfrac{%
\partial ^{2}f}{\partial t\partial s}\left( b,c\right) \right \vert
^{q}+\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s}\left(
b,d\right) \right \vert ^{q}}{4}\right] ^{\frac{1}{q}}  \notag
\end{eqnarray}%
where $\frac{1}{p}+\frac{1}{q}=1$ and $H(t,s)$ defined by%
\begin{equation*}
H(t,s)=\left \vert \int \limits_{U_{1}(t)}^{U_{2}(t)}\int
\limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right \vert
\end{equation*}%
with $U_{1}(t)=(1-t)a+tb,$ $U_{2}(t)=ta+(1-t)b,$ $V_{1}(s)=(1-s)c+sd$ and $%
V_{2}(s)=sc+(1-s)d.$
\end{theorem}

\begin{proof}
Taking the modulus in Lemma \ref{l2} and by using the well-known H\"{o}%
lder's inequality, we have%
\begin{eqnarray}
&&\left \vert \frac{f(a,c)+f(a,d)+f(b,c)+f(b,d)}{4}\int
\limits_{a}^{b}\int \limits_{c}^{d}p(x,y)dydx\right.  \label{e29} \\
&&  \notag \\
&&\left. -\frac{1}{4}\int \limits_{a}^{b}\int \limits_{c}^{d}\left[
f(x,c)+f(x,d)+f(a,y)+f(b,y)\right] p(x,y)dydx+\int \limits_{a}^{b}\int
\limits_{c}^{d}f(x,y)p(x,y)dydx\right \vert  \notag \\
&&  \notag \\
&=&\frac{\left( b-a\right) \left( d-c\right) }{4}\left \vert \int
\limits_{0}^{1}\int \limits_{0}^{1}\left[ \int%
\limits_{U_{1}(t)}^{U_{2}(t)}\int \limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv%
\right] \dfrac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) dsdt\right \vert  \notag \\
&&  \notag \\
&\leq &\frac{\left( b-a\right) \left( d-c\right) }{4}\int
\limits_{0}^{1}\int \limits_{0}^{1}\left \vert \int
\limits_{U_{1}(t)}^{U_{2}(t)}\int
\limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right \vert \left \vert \dfrac{%
\partial ^{2}f}{\partial t\partial s}\left( U_{2}(t),V_{2}(s)\right) \right
\vert dsdt  \notag \\
&&  \notag \\
&\leq &\frac{\left( b-a\right) \left( d-c\right) }{4}\left( \int
\limits_{0}^{1}\int \limits_{0}^{1}\left \vert \int
\limits_{U_{1}(t)}^{U_{2}(t)}\int
\limits_{V_{1}(s)}^{V_{2}(s)}p(u,v)dudv\right \vert ^{p}dsdt\right) ^{\frac{1%
}{p}}\left( \int \limits_{0}^{1}\int \limits_{0}^{1}\left \vert \dfrac{%
\partial ^{2}f}{\partial t\partial s}\left( U_{2}(t),V_{2}(s)\right) \right
\vert ^{q}dsdt\right) ^{\frac{1}{q}}.  \notag
\end{eqnarray}%
Since $\left \vert \dfrac{\partial ^{2}f}{\partial t\partial s}\right \vert
^{q},$ $q>1,$ is a co-ordinated convex function on $\Delta ,$ we obtain%
\begin{eqnarray}
&&\int \limits_{0}^{1}\int \limits_{0}^{1}\left \vert \dfrac{\partial ^{2}f%
}{\partial t\partial s}\left( U_{2}(t),V_{2}(s)\right) \right \vert ^{q}dsdt
\label{e28} \\
&=&\int \limits_{0}^{1}\int \limits_{0}^{1}\left \vert \dfrac{\partial
^{2}f}{\partial t\partial s}\left( ta+(1-t)b,sa+(1-s)d\right) \right \vert
^{q}dsdt  \notag \\
&&  \notag \\
&\leq &\int \limits_{0}^{1}\int \limits_{0}^{1}\left[ ts\left \vert \tfrac{%
\partial ^{2}f}{\partial t\partial s}\left( a,c\right) \right \vert
^{q}+t(1-s)\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s}\left(
a,d\right) \right \vert ^{q}\right.  \notag \\
&&  \notag \\
&&\left. +(1-t)s\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s}%
\left( b,c\right) \right \vert ^{q}+(1-t)(1-s)\left \vert \tfrac{\partial
^{2}f}{\partial t\partial s}\left( b,d\right) \right \vert ^{q}\right] dsdt
\notag \\
&&  \notag \\
&=&\frac{\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s}\left(
a,c\right) \right \vert ^{q}+\left \vert \tfrac{\partial ^{2}f}{\partial
t\partial s}\left( a,d\right) \right \vert ^{q}+\left \vert \tfrac{\partial
^{2}f}{\partial t\partial s}\left( b,c\right) \right \vert ^{q}+\left \vert
\tfrac{\partial ^{2}f}{\partial t\partial s}\left( b,d\right) \right \vert
^{q}}{4}.  \notag
\end{eqnarray}%
If we substitute the inequality (\ref{e28}) in (\ref{e29}), the we establish
desired result.
\end{proof}

\begin{remark}
If we choose $p(x,y)=1,$ then we obtain the following inequality%
\begin{eqnarray*}
&&\left \vert \frac{f(a,c)+f(a,d)+f(b,c)+f(b,d)}{4}\right. \\
&& \\
&&-\frac{1}{4}\left[ \frac{1}{b-a}\int \limits_{a}^{b}f(x,c)dx+\frac{1}{b-a}%
\int \limits_{a}^{b}f(x,d)dx\frac{1}{d-c}\int \limits_{c}^{d}f(a,y)dy+\frac{1%
}{d-c}\int \limits_{c}^{d}f(b,y)dy\right] \\
&& \\
&&\left. +\frac{1}{\left( b-a\right) \left( d-c\right) }\int
\limits_{a}^{b}\int \limits_{c}^{d}f(x,y)dydx\right \vert \\
&& \\
&\leq &\frac{\left( b-a\right) \left( d-c\right) }{4(p+1)^{\frac{2}{p}}}%
\left[ \frac{\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s}\left(
a,c\right) \right \vert ^{q}+\left \vert \tfrac{\partial ^{2}f}{\partial
t\partial s}\left( a,d\right) \right \vert ^{q}+\left \vert \tfrac{\partial
^{2}f}{\partial t\partial s}\left( b,c\right) \right \vert ^{q}+\left \vert
\tfrac{\partial ^{2}f}{\partial t\partial s}\left( b,d\right) \right \vert
^{q}}{4}\right] ^{\frac{1}{q}}
\end{eqnarray*}%
which was proved by Sarikaya et al. in \cite{sarikaya}.
\end{remark}

\section{Weighted Simpson Type Inequalities}

In this section, we first define the following mapping%
\begin{eqnarray*}
&&\Theta (a,b;f,p) \\
&=&f\left( \frac{a+b}{2},\frac{c+d}{2}\right) \int_{\frac{1}{6}}^{\frac{5}{6}%
}\int_{\frac{1}{6}}^{\frac{5}{6}}p\left( u,v\right) dvdu \\
&&+f\left( a,\frac{c+d}{2}\right) \int_{\frac{5}{6}}^{1}\int_{\frac{1}{6}}^{%
\frac{5}{6}}p\left( u,v\right) dvdu+f\left( b,\frac{c+d}{2}\right) \int_{0}^{%
\frac{1}{6}}\int_{\frac{1}{6}}^{\frac{5}{6}}p\left( u,v\right) dvdu \\
&&+f\left( \frac{a+b}{2},c\right) \int_{\frac{1}{6}}^{\frac{5}{6}}\int_{%
\frac{5}{6}}^{1}p\left( u,v\right) dvdu+f\left( \frac{a+b}{2},d\right) \int_{%
\frac{1}{6}}^{\frac{5}{6}}\int_{0}^{\frac{1}{6}}p\left( u,v\right) dvdu \\
&&+f\left( a,c\right) \int_{\frac{5}{6}}^{1}\int_{\frac{5}{6}}^{1}p\left(
u,v\right) dvdu+f\left( b,c\right) \int_{0}^{\frac{1}{6}}\int_{\frac{5}{6}%
}^{1}p\left( u,v\right) dvdu \\
&&+f\left( a,d\right) \int_{\frac{5}{6}}^{1}\int_{0}^{\frac{1}{6}}p\left(
u,v\right) dvdu+f\left( b,d\right) \int_{0}^{\frac{1}{6}}\int_{0}^{\frac{1}{6%
}}p\left( u,v\right) dvdu \\
&&-\frac{1}{b-a}\int_{a}^{b}\left( \int_{\frac{5}{6}}^{1}p\left( \frac{b-x}{%
b-a},v\right) dv\right) f\left( x,c\right) dx-\frac{1}{b-a}%
\int_{a}^{b}\left( \int_{\frac{1}{2}}^{\frac{5}{6}}p\left( \frac{b-x}{b-a}%
,v\right) dv\right) f\left( x,\frac{c+d}{2}\right) dx \\
&&-\frac{1}{b-a}\int_{a}^{b}\left( \int_{0}^{\frac{1}{6}}p\left( \frac{b-x}{%
b-a},v\right) dv\right) f\left( x,d\right) dx-\frac{1}{d-c}%
\int_{c}^{d}\left( \int_{\frac{5}{6}}^{1}p\left( u,\frac{d-y}{d-c}\right)
du\right) f\left( a,y\right) dy \\
&&-\frac{1}{d-c}\int_{c}^{d}\left( \int_{\frac{1}{2}}^{\frac{5}{6}}p\left( u,%
\frac{d-y}{d-c}\right) du\right) f\left( \frac{a+b}{2},y\right) dy-\frac{1}{%
d-c}\int_{c}^{d}\left( \int_{0}^{\frac{1}{6}}p\left( u,\frac{d-y}{d-c}%
\right) du\right) f\left( b,y\right) dy \\
&&+\frac{1}{\left( b-a\right) \left( d-c\right) }\int_{a}^{b}\int_{c}^{d}p%
\left( \frac{b-x}{b-a},\frac{d-y}{d-c}\right) f\left( x,y\right) dydx.
\end{eqnarray*}%
Now, we give the following equality.

\begin{lemma}
\label{l3} Let the mappings $p,$ $U_{2}$ and $V_{2}$ be as in Lemma \ref{l2}
and let $f:\Delta :=\left[ a,b\right] \times \left[ c,d\right] \rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ be a twice partially differentiable mapping on $\Delta ^{\circ }$. If $%
\dfrac{\partial ^{2}f}{\partial t\partial s}\in L(\Delta ),$ then we have
the following equality%
\begin{equation*}
\Theta (a,b;f,p)=\left( b-a\right) \left( d-c\right)
\int_{0}^{1}\int_{0}^{1}w\left( t,s\right) \frac{\partial ^{2}f}{\partial
t\partial s}\left( U_{2}(t),V_{2}(s)\right) dsdt
\end{equation*}%
where the mapping $w:\left[ 0,1\right] \times \left[ 0,1\right] \rightarrow
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
$ is defined by%
\begin{equation*}
w\left( t,s\right) =\left \{
\begin{array}{ccc}
\int_{\frac{1}{6}}^{t}\int_{\frac{1}{6}}^{s}p\left( u,v\right) dudv &
0\leq t\leq \frac{1}{2}, & 0\leq s\leq \frac{1}{2} \\
\int_{\frac{1}{6}}^{t}\int_{\frac{5}{6}}^{s}p\left( u,v\right) dudv &
0\leq t\leq \frac{1}{2}, & \frac{1}{2}\leq s\leq 1 \\
\int_{\frac{5}{6}}^{t}\int_{\frac{1}{6}}^{s}p\left( u,v\right) dudv &
\frac{1}{2}\leq t\leq 1, & 0\leq s\leq \frac{1}{2} \\
\int_{\frac{5}{6}}^{t}\int_{\frac{5}{6}}^{s}p\left( u,v\right) dudv &
\frac{1}{2}\leq t\leq 1, & \frac{1}{2}\leq s\leq 1.%
\end{array}%
\right.
\end{equation*}
\end{lemma}

\begin{proof}
From the definition of the mapping $w$, we have%
\begin{eqnarray*}
&&\left( b-a\right) \left( d-c\right) \int_{0}^{1}\int_{0}^{1}w\left(
t,s\right) \frac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) dsdt \\
&=&\left( b-a\right) \left( d-c\right) \int_{0}^{\frac{1}{2}}\int_{0}^{\frac{%
1}{2}}\left( \int_{\frac{1}{6}}^{t}\int_{\frac{1}{6}}^{s}p\left( u,v\right)
dvdu\right) \frac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) dtds \\
&&+\left( b-a\right) \left( d-c\right) \int_{0}^{\frac{1}{2}}\int_{\frac{1}{2%
}}^{1}\left( \int_{\frac{1}{6}}^{t}\int_{\frac{5}{6}}^{s}p\left( u,v\right)
dudv\right) \frac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) dtds \\
&&+\left( b-a\right) \left( d-c\right) \int_{\frac{1}{2}}^{1}\int_{0}^{\frac{%
1}{2}}\left( \int_{\frac{5}{6}}^{t}\int_{\frac{1}{6}}^{s}p\left( u,v\right)
dudv\right) \frac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) dtds \\
&&+\left( b-a\right) \left( d-c\right) \int_{\frac{1}{2}}^{1}\int_{\frac{1}{2%
}}^{1}\left( \int_{\frac{5}{6}}^{t}\int_{\frac{5}{6}}^{s}p\left( u,v\right)
dudv\right) \frac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) dtds \\
&=&\left( b-a\right) \left( d-c\right) \left[ J_{1}+J_{2}+J_{3}+J_{4}\right]
.
\end{eqnarray*}%
Integration by parts, we obtain%
\begin{eqnarray*}
J_{1} &=&\int_{0}^{\frac{1}{2}}\int_{0}^{\frac{1}{2}}\left( \int_{\frac{1}{6}%
}^{t}\int_{\frac{1}{6}}^{s}p\left( u,v\right) dvdu\right) \frac{\partial
^{2}f}{\partial t\partial s}\left( U_{2}(t),V_{2}(s)\right) dtds \\
&=&\frac{1}{c-d}\int_{0}^{\frac{1}{2}}\left( \int_{\frac{1}{6}}^{t}\int_{%
\frac{1}{6}}^{\frac{1}{2}}p\left( u,v\right) dvdu\right) \frac{\partial f}{%
\partial t}\left( U_{2}(t),\frac{c+d}{2}\right) dt \\
&&-\frac{1}{c-d}\int_{0}^{\frac{1}{2}}\left( \int_{\frac{1}{6}}^{t}\int_{%
\frac{1}{6}}^{0}p\left( u,v\right) dvdu\right) \frac{\partial f}{\partial t}%
\left( U_{2}(t),d\right) dt \\
&&\left. -\frac{1}{c-d}\int_{0}^{\frac{1}{2}}\int_{0}^{\frac{1}{2}}\left(
\int_{\frac{1}{6}}^{t}p\left( s,v\right) dv\right) \frac{\partial f}{%
\partial t}\left( U_{2}(t),V_{2}(s)\right) dtds\right]  \\
&=&\frac{1}{c-d}\left[ \frac{1}{a-b}\left( \int_{\frac{1}{6}}^{\frac{1}{2}%
}\int_{\frac{1}{6}}^{\frac{1}{2}}p\left( u,v\right) dvdu\right) f\left(
\frac{a+b}{2},\frac{c+d}{2}\right) \right.  \\
&&\left. -\frac{1}{a-b}\left( \int_{\frac{1}{6}}^{0}\int_{\frac{1}{6}}^{%
\frac{1}{2}}p\left( u,v\right) dvdu\right) f\left( b,\frac{c+d}{2}\right)
\right.  \\
&&\left. -\frac{1}{a-b}\int_{0}^{\frac{1}{2}}\left( \int_{\frac{1}{6}}^{%
\frac{1}{2}}p\left( t,v\right) dv\right) f\left( U_{2}(t),\frac{c+d}{2}%
\right) dt\right]  \\
&&-\frac{1}{c-d}\left[ \frac{1}{a-b}\left( \int_{\frac{1}{6}}^{\frac{1}{2}%
}\int_{\frac{1}{6}}^{0}p\left( u,v\right) dvdu\right) f\left( \frac{a+b}{2}%
,d\right) \right.  \\
&&\left. -\frac{1}{a-b}\left( \int_{\frac{1}{6}}^{0}\int_{\frac{1}{6}%
}^{0}p\left( u,v\right) dvdu\right) f\left( b,d\right) \right.  \\
&&\left. -\frac{1}{a-b}\int_{0}^{\frac{1}{2}}\left( \int_{\frac{1}{6}%
}^{0}p\left( t,v\right) dv\right) f\left( U_{2}(t),d\right) dt\right]  \\
&&-\left[ \frac{1}{c-d}\int_{0}^{\frac{1}{2}}\frac{1}{a-b}\left( \int_{\frac{%
1}{6}}^{\frac{1}{2}}p\left( s,v\right) dv\right) f\left( \frac{a+b}{2}%
,V_{2}(s)\right) \right.  \\
&&\left. -\frac{1}{a-b}\left( \int_{\frac{1}{6}}^{0}\int_{\frac{1}{6}%
}^{0}p\left( u,v\right) dvdu\right) f\left( b,d\right) \right.  \\
&&\left. -\frac{1}{a-b}\int_{0}^{\frac{1}{2}}p\left( s,t\right) f\left(
U_{2}(t),V_{2}(s)\right) dt\right] ds
\end{eqnarray*}%
\begin{eqnarray*}
&=&\frac{1}{\left( b-a\right) \left( d-c\right) }\left[ f\left( \frac{a+b}{2}%
,\frac{c+d}{2}\right) \int_{\frac{1}{6}}^{\frac{1}{2}}\int_{\frac{1}{6}%
}^{1}p\left( u,v\right) dvdu+f\left( b,\frac{c+d}{2}\right) \int_{0}^{\frac{1%
}{6}}\int_{\frac{5}{6}}^{1}p\left( u,v\right) dvdu\right.  \\
&&-\int_{0}^{\frac{1}{2}}\left( \int_{\frac{5}{6}}^{\frac{1}{2}}p\left(
t,v\right) dv\right) f\left( U_{2}(t),\frac{c+d}{2}\right) dt+f\left( \frac{%
a+b}{2},d\right) \int_{\frac{1}{6}}^{\frac{1}{2}}\int_{0}^{\frac{1}{6}%
}p\left( u,v\right) dudv \\
&&+f\left( b,d\right) \int_{0}^{\frac{1}{6}}\int_{\frac{1}{2}}^{\frac{1}{6}%
}p\left( u,v\right) dudv-\int_{0}^{\frac{1}{2}}\left( \int_{0}^{\frac{1}{6}%
}p\left( t,v\right) dv\right) f\left( U_{2}(t),d\right) dt \\
&&-\int_{0}^{\frac{1}{2}}\left( \int_{\frac{1}{6}}^{\frac{1}{2}}p\left(
s,v\right) dv\right) f\left( \frac{a+b}{2},V_{2}(s)\right) ds-\int_{0}^{%
\frac{1}{2}}\left( \int_{0}^{\frac{1}{6}}p\left( s,v\right) dv\right)
f\left( b,V_{2}(s)\right) ds \\
&&\left. +\int_{0}^{\frac{1}{2}}\int_{0}^{\frac{1}{2}}p\left( s,t\right)
f\left( U_{2}(t),V_{2}(s)\right) dtds\right] .
\end{eqnarray*}%
Similarly, we get%
\begin{eqnarray*}
J_{2} &=&\int_{0}^{\frac{1}{2}}\int_{\frac{1}{2}}^{1}\left( \int_{\frac{1}{6}%
}^{t}\int_{\frac{5}{6}}^{s}p\left( u,v\right) dudv\right) \frac{\partial
^{2}f}{\partial t\partial s}\left( U_{2}(t),V_{2}(s)\right) dtds \\
&=&\frac{1}{\left( b-a\right) \left( d-c\right) }\left[ f\left( \frac{a+b}{2}%
,c\right) \int_{\frac{1}{6}}^{\frac{1}{2}}\int_{\frac{5}{6}}^{1}p\left(
u,v\right) dudv+f\left( b,c\right) \int_{0}^{\frac{1}{6}}\int_{\frac{5}{6}%
}^{1}p\left( u,v\right) dudv\right.  \\
&&-\int_{0}^{\frac{1}{2}}\left( \int_{\frac{5}{6}}^{1}p\left( t,v\right)
dv\right) f\left( at+\left( 1-t\right) b,c\right) dt+f\left( \frac{a+b}{2},%
\frac{c+d}{2}\right) \int_{\frac{1}{6}}^{\frac{1}{2}}\int_{\frac{1}{2}}^{%
\frac{5}{6}}p\left( u,v\right) dudv \\
&&+f\left( b,\frac{c+d}{2}\right) \int_{0}^{\frac{1}{6}}\int_{\frac{1}{2}}^{%
\frac{5}{6}}p\left( u,v\right) dudv-\int_{0}^{\frac{1}{2}}\left( \int_{\frac{%
1}{2}}^{\frac{5}{6}}p\left( t,v\right) dv\right) f\left( U_{2}(t),\frac{c+d}{%
2}\right) dt \\
&&-\int_{\frac{1}{2}}^{1}\left( \int_{\frac{1}{6}}^{\frac{1}{2}}p\left(
u,s\right) ds\right) f\left( \frac{a+b}{2},V_{2}(s)\right) ds-\int_{\frac{1}{%
2}}^{1}\left( \int_{0}^{\frac{1}{6}}p\left( u,s\right) ds\right) f\left(
b,V_{2}(s)\right) ds \\
&&\left. +\int_{\frac{1}{2}}^{1}\int_{0}^{\frac{1}{2}}p\left( t,s\right)
f\left( U_{2}(t),V_{2}(s)\right) dtds\right] ,
\end{eqnarray*}%
\begin{eqnarray*}
J_{3} &=&\int_{\frac{1}{2}}^{1}\int_{0}^{\frac{1}{2}}\left( \int_{\frac{5}{6}%
}^{t}\int_{\frac{1}{6}}^{s}p\left( u,v\right) dudv\right) \frac{\partial
^{2}f}{\partial t\partial s}\left( U_{2}(t),V_{2}(s)\right) dtds \\
&=&\frac{1}{\left( b-a\right) \left( d-c\right) }f\left[ \left( a,\frac{c+d}{%
2}\right) \int_{\frac{5}{6}}^{1}\int_{\frac{1}{6}}^{\frac{1}{2}}p\left(
u,v\right) dvdu+f\left( \frac{a+b}{2},\frac{c+d}{2}\right) \int_{\frac{1}{2}%
}^{\frac{5}{6}}\int_{\frac{1}{6}}^{\frac{1}{2}}p\left( u,v\right)
dvdu\right.  \\
&&-\int_{\frac{1}{2}}^{1}\left( \int_{\frac{1}{6}}^{\frac{1}{2}}p\left(
t,v\right) dv\right) f\left( U_{2}(t),\frac{c+d}{2}\right) dt+f\left(
a,d\right) \int_{\frac{5}{6}}^{1}\int_{0}^{\frac{1}{6}}p\left( u,v\right)
dvdu \\
&&+f\left( \frac{a+b}{2},d\right) \int_{\frac{1}{2}}^{\frac{5}{6}}\int_{0}^{%
\frac{1}{6}}p\left( u,v\right) dvdu-\int_{\frac{1}{2}}^{1}\left( \int_{0}^{%
\frac{1}{6}}p\left( t,v\right) dv\right) f\left( U_{2}(t),d\right) dt \\
&&-\int_{0}^{\frac{1}{2}}\left( \int_{\frac{5}{6}}^{1}p\left( u,s\right)
du\right) f\left( a,V_{2}(s)\right) ds-\int_{0}^{\frac{1}{2}}\left( \int_{%
\frac{1}{2}}^{\frac{5}{6}}p\left( u,s\right) du\right) f\left( \frac{a+b}{2}%
,V_{2}(s)\right) ds \\
&&\left. +\int_{0}^{\frac{1}{2}}\int_{\frac{1}{2}}^{1}p\left( t,s\right)
f\left( U_{2}(t),V_{2}(s)\right) dtds\right] ,
\end{eqnarray*}%
and%
\begin{eqnarray*}
J_{4} &=&\int_{\frac{1}{2}}^{1}\int_{\frac{1}{2}}^{1}\left( \int_{\frac{5}{6}%
}^{t}\int_{\frac{5}{6}}^{s}p\left( u,v\right) dudv\right) \frac{\partial
^{2}f}{\partial t\partial s}\left( U_{2}(t),V_{2}(s)\right) dtds \\
&=&\frac{1}{\left( b-a\right) \left( d-c\right) }\left[ f\left( a,c\right)
\int_{\frac{5}{6}}^{\frac{1}{2}}\int_{\frac{5}{6}}^{1}p\left( u,v\right)
dvdu+f\left( \frac{a+b}{2},c\right) \int_{\frac{1}{2}}^{\frac{5}{6}}\int_{%
\frac{5}{6}}^{1}p\left( u,v\right) dvdu\right.  \\
&&-\int_{\frac{1}{2}}^{1}\left( \int_{\frac{5}{6}}^{1}p\left( t,v\right)
dv\right) f\left( U_{2}(t),c\right) dt+f\left( a,\frac{c+d}{2}\right) \int_{%
\frac{5}{6}}^{1}\int_{\frac{1}{2}}^{\frac{5}{6}}p\left( u,v\right) dvdu \\
&&+f\left( \frac{a+b}{2},\frac{c+d}{2}\right) \int_{\frac{1}{2}}^{\frac{5}{6}%
}\int_{\frac{1}{2}}^{\frac{5}{6}}p\left( u,v\right) dvdu-\int_{\frac{1}{2}%
}^{1}\left( \int_{\frac{1}{2}}^{\frac{5}{6}}p\left( t,v\right) dv\right)
f\left( U_{2}(t),\frac{c+d}{2}\right) dt \\
&&-\int_{\frac{1}{2}}^{1}\left( \int_{\frac{5}{6}}^{1}p\left( u,s\right)
du\right) f\left( a,V_{2}(s)\right) ds-\int_{\frac{1}{2}}^{1}\left( \int_{%
\frac{1}{2}}^{\frac{5}{6}}p\left( u,s\right) du\right) f\left( \frac{a+b}{2}%
,V_{2}(s)\right) ds \\
&&\left. +\int_{\frac{1}{2}}^{1}\int_{\frac{1}{2}}^{1}p\left( t,s\right)
f\left( U_{2}(t),V_{2}(s)\right) dtds\right] .
\end{eqnarray*}%
Using the change of variable, we obtain%
\begin{eqnarray*}
&&\left( b-a\right) \left( d-c\right) \left[ J_{1}+J_{2}+J_{3}+J_{4}\right]
\\
&=&f\left( \frac{a+b}{2},\frac{c+d}{2}\right) \int_{\frac{1}{6}}^{\frac{5}{6}%
}\int_{\frac{1}{6}}^{\frac{5}{6}}p\left( u,v\right) dvdu \\
&&+f\left( a,\frac{c+d}{2}\right) \int_{\frac{5}{6}}^{1}\int_{\frac{1}{6}}^{%
\frac{5}{6}}p\left( u,v\right) dvdu+f\left( b,\frac{c+d}{2}\right) \int_{0}^{%
\frac{1}{6}}\int_{\frac{1}{6}}^{\frac{5}{6}}p\left( u,v\right) dvdu \\
&&+f\left( \frac{a+b}{2},c\right) \int_{\frac{1}{6}}^{\frac{5}{6}}\int_{%
\frac{5}{6}}^{1}p\left( u,v\right) dvdu+f\left( \frac{a+b}{2},d\right) \int_{%
\frac{1}{6}}^{\frac{5}{6}}\int_{0}^{\frac{1}{6}}p\left( u,v\right) dvdu \\
&&+f\left( a,c\right) \int_{\frac{5}{6}}^{1}\int_{\frac{5}{6}}^{1}p\left(
u,v\right) dvdu+f\left( b,c\right) \int_{0}^{\frac{1}{6}}\int_{\frac{5}{6}%
}^{1}p\left( u,v\right) dvdu \\
&&+f\left( a,d\right) \int_{\frac{5}{6}}^{1}\int_{0}^{\frac{1}{6}}p\left(
u,v\right) dvdu+f\left( b,d\right) \int_{0}^{\frac{1}{6}}\int_{0}^{\frac{1}{6%
}}p\left( u,v\right) dvdu \\
&&-\frac{1}{b-a}\int_{a}^{b}\left( \int_{\frac{5}{6}}^{1}p\left( \frac{b-x}{%
b-a},v\right) dv\right) f\left( x,c\right) dx-\frac{1}{b-a}%
\int_{a}^{b}\left( \int_{\frac{1}{2}}^{\frac{5}{6}}p\left( \frac{b-x}{b-a}%
,v\right) dv\right) f\left( x,\frac{c+d}{2}\right) dx \\
&&-\frac{1}{b-a}\int_{a}^{b}\left( \int_{0}^{\frac{1}{6}}p\left( \frac{b-x}{%
b-a},v\right) dv\right) f\left( x,d\right) dx-\frac{1}{d-c}%
\int_{c}^{d}\left( \int_{\frac{5}{6}}^{1}p\left( u,\frac{d-y}{d-c}\right)
du\right) f\left( a,y\right) dy \\
&&-\frac{1}{d-c}\int_{c}^{d}\left( \int_{\frac{1}{2}}^{\frac{5}{6}}p\left( u,%
\frac{d-y}{d-c}\right) du\right) f\left( \frac{a+b}{2},y\right) dy-\frac{1}{%
d-c}\int_{c}^{d}\left( \int_{0}^{\frac{1}{6}}p\left( u,\frac{d-y}{d-c}%
\right) du\right) f\left( b,y\right) dy \\
&&+\frac{1}{\left( b-a\right) \left( d-c\right) }\int_{a}^{b}\int_{c}^{d}p%
\left( \frac{b-x}{b-a},\frac{d-y}{d-c}\right) f\left( x,y\right) dydx
\end{eqnarray*}%
which completes the proof.
\end{proof}

\begin{remark}
If we choose $p(x,y)=1$ in Lemma \ref{l3}, then Lemma \ref{l3} reduses to
Lemma \ref{l1} which is proved by \cite{ozdemir2}.
\end{remark}

\begin{theorem}
\label{t2} Let the mappings $p,$ $U_{2}$ and $V_{2}$ be as in Lemma \ref{l2}%
. If $\left \vert \dfrac{\partial ^{2}f}{\partial t\partial s}\right \vert
^{q},$ $q>1,$ is a co-ordinated convex function on $\Delta ,$ then we have
the following inequality%
\begin{eqnarray*}
\left \vert \Theta (a,b;f,p)\right \vert &\leq &\left( b-a\right) \left(
d-c\right) \left( \int_{0}^{1}\int_{0}^{1}\left \vert w\left( t,s\right)
\right \vert ^{p}dsdt\right) ^{\frac{1}{p}} \\
&&\times \left( \frac{\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s%
}\left( a,c\right) \right \vert ^{q}+\left \vert \tfrac{\partial ^{2}f}{%
\partial t\partial s}\left( a,d\right) \right \vert ^{q}+\left \vert \tfrac{%
\partial ^{2}f}{\partial t\partial s}\left( b,c\right) \right \vert
^{q}+\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s}\left(
b,d\right) \right \vert ^{q}}{4}\right) ^{\frac{1}{q}}
\end{eqnarray*}%
where $\frac{1}{p}+\frac{1}{q}=1$.
\end{theorem}

\begin{proof}
Taking the modulus in Lemma \ref{l3} and using the H\"{o}lders inequality we
have%
\begin{eqnarray*}
\left \vert \Theta (a,b;f,p)\right \vert &\leq &\left( b-a\right) \left(
d-c\right) \int_{0}^{1}\int_{0}^{1}\left \vert w\left( t,s\right) \right
\vert \left \vert \frac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) \right \vert dsdt \\
&& \\
&\leq &\left( b-a\right) \left( d-c\right) \left(
\int_{0}^{1}\int_{0}^{1}\left \vert w\left( t,s\right) \right \vert
^{p}dsdt\right) ^{\frac{1}{p}}\left( \int_{0}^{1}\int_{0}^{1}\left \vert
\frac{\partial ^{2}f}{\partial t\partial s}\left( U_{2}(t),V_{2}(s)\right)
\right \vert ^{q}dsdt\right) ^{\frac{1}{q}}.
\end{eqnarray*}%
Since $\left \vert \dfrac{\partial ^{2}f}{\partial t\partial s}\right \vert
^{q},$ $q>1,$ is a co-ordinated convex function on $\Delta ,$ using the
inequality (\ref{e28}), we obtain%
\begin{eqnarray*}
\left \vert \Theta (a,b;f,p)\right \vert &\leq &\left( b-a\right) \left(
d-c\right) \left( \int_{0}^{1}\int_{0}^{1}\left \vert w\left( t,s\right)
\right \vert ^{p}dsdt\right) ^{\frac{1}{p}} \\
&&\times \left( \frac{\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s%
}\left( a,c\right) \right \vert ^{q}+\left \vert \tfrac{\partial ^{2}f}{%
\partial t\partial s}\left( a,d\right) \right \vert ^{q}+\left \vert \tfrac{%
\partial ^{2}f}{\partial t\partial s}\left( b,c\right) \right \vert
^{q}+\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s}\left(
b,d\right) \right \vert ^{q}}{4}\right) ^{\frac{1}{q}}.
\end{eqnarray*}%
This completes the proof.
\end{proof}

\begin{corollary}
If we choose $p(x,y)=1$ in Theorem \ref{t2}, then we have the following
inequality%
\begin{eqnarray*}
&&\left \vert \frac{f\left( a,\frac{c+d}{2}\right) +f\left( b,\frac{c+d}{2}%
\right) +4f\left( \frac{a+b}{2},\frac{c+d}{2}\right) +f\left( \frac{a+b}{2}%
,c\right) +f\left( \frac{a+b}{2},d\right) }{9}\right. \\
&& \\
&&+\frac{f\left( a,c\right) +f\left( b,c\right) +f\left( a,d\right) +f\left(
b,d\right) }{36}-\frac{1}{6\left( b-a\right) }\int_{a}^{b}\left[ f\left(
x,c\right) +4f\left( x,\frac{c+d}{2}\right) +f\left( x,d\right) \right] dx \\
&& \\
&&\left. -\frac{1}{6\left( d-c\right) }\int_{c}^{d}\left[ f\left( a,y\right)
+4f\left( \frac{a+b}{2},y\right) +f\left( b,y\right) \right] dy+\frac{1}{%
\left( b-a\right) \left( d-c\right) }\int_{a}^{b}\int_{c}^{d}f\left(
x,y\right) dydx\right \vert \\
&& \\
&\leq &\frac{\left( b-a\right) \left( d-c\right) }{36}\left[ 2^{p+1}+1\right]
^{\frac{2}{p}}\left( \frac{1}{9\left( p+1\right) ^{2}}\right) ^{\frac{1}{p}}
\\
&&\times \left( \frac{\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s%
}\left( a,c\right) \right \vert ^{q}+\left \vert \tfrac{\partial ^{2}f}{%
\partial t\partial s}\left( a,d\right) \right \vert ^{q}+\left \vert \tfrac{%
\partial ^{2}f}{\partial t\partial s}\left( b,c\right) \right \vert
^{q}+\left \vert \tfrac{\partial ^{2}f}{\partial t\partial s}\left(
b,d\right) \right \vert ^{q}}{4}\right) ^{\frac{1}{q}}.
\end{eqnarray*}
\end{corollary}

\begin{theorem}
\label{t3} Let the mappings $p,$ $U_{2}$ and $V_{2}$ be as in Lemma \ref{l2}%
. If $\dfrac{\partial ^{2}f}{\partial t\partial s}$ is bounded on $\Delta ,$
i.e.%
\begin{equation*}
\left \Vert \dfrac{\partial ^{2}f}{\partial t\partial s}\right \Vert
_{\infty }=\sup_{\left( x,y\right) \in \Delta }\left \vert \dfrac{\partial
^{2}f}{\partial t\partial s}(x,y)\right \vert <\infty ,
\end{equation*}%
then we have the following inequality%
\begin{equation*}
\left \vert \Theta (a,b;f,p)\right \vert \leq \left( b-a\right) \left(
d-c\right) \left \Vert \dfrac{\partial ^{2}f}{\partial t\partial s}\right
\Vert _{\infty }\int_{0}^{1}\int_{0}^{1}\left \vert w\left( t,s\right)
\right \vert dsdt.
\end{equation*}
\end{theorem}

\begin{proof}
From Lemma \ref{l3}, we have%
\begin{equation*}
\left \vert \Theta (a,b;f,p)\right \vert \leq \left( b-a\right) \left(
d-c\right) \int_{0}^{1}\int_{0}^{1}\left \vert w\left( t,s\right) \right \vert
\left \vert \frac{\partial ^{2}f}{\partial t\partial s}\left(
U_{2}(t),V_{2}(s)\right) \right \vert dsdt.
\end{equation*}%
Since $\dfrac{\partial ^{2}f}{\partial t\partial s}$ is bounded on $\Delta ,$
we obtain%
\begin{equation*}
\left \vert \Theta (a,b;f,p)\right \vert \leq \left( b-a\right) \left(
d-c\right) \left \Vert \dfrac{\partial ^{2}f}{\partial t\partial s}%
\right \Vert _{\infty }\int_{0}^{1}\int_{0}^{1}\left \vert w\left( t,s\right)
\right \vert dsdt
\end{equation*}%
which completes the proof.
\end{proof}

\begin{remark}
If we choose $p(x,y)=1$ in Theorem \ref{t3}, then we have the following
inequality%
\begin{eqnarray*}
&&\left \vert \frac{f\left( a,\frac{c+d}{2}\right) +f\left( b,\frac{c+d}{2}%
\right) +4f\left( \frac{a+b}{2},\frac{c+d}{2}\right) +f\left( \frac{a+b}{2}%
,c\right) +f\left( \frac{a+b}{2},d\right) }{9}\right. \\
&& \\
&&+\frac{f\left( a,c\right) +f\left( b,c\right) +f\left( a,d\right) +f\left(
b,d\right) }{36}-\frac{1}{6\left( b-a\right) }\int_{a}^{b}\left[ f\left(
x,c\right) +4f\left( x,\frac{c+d}{2}\right) +f\left( x,d\right) \right] dx \\
&& \\
&&\left. -\frac{1}{6\left( d-c\right) }\int_{c}^{d}\left[ f\left( a,y\right)
+4f\left( \frac{a+b}{2},y\right) +f\left( b,y\right) \right] dy+\frac{1}{%
\left( b-a\right) \left( d-c\right) }\int_{a}^{b}\int_{c}^{d}f\left(
x,y\right) dydx\right \vert \\
&& \\
&\leq &\frac{25\left( b-a\right) \left( d-c\right) }{1296}\left \Vert \dfrac{%
\partial ^{2}f}{\partial t\partial s}\right \Vert _{\infty }
\end{eqnarray*}%
which was proved by \"{O}zdemir et al. in \cite{ozdemir2}.
\end{remark}



\end{document}
